Author's response to reviews

Title: Success with antiretroviral treatment for children in Kigali, Rwanda: experience with health center/nurse-based care

Authors:

Johan van Griensven (jvgrie@yahoo.com)
Ludwig De Naeyer (interact-ta@terramail.rw)
Jeanine Uwera (jinine06@yahoo.fr)
Anita Asiimwe (anitaa@tracrwanda.org)
Claire Gazille (clairugema@yahoo.fr)
Tony Reid (tony.reid@brussels.msf.org)

Version: 6 Date: 28 September 2008

Author's response to reviews: see over
Dear Editor,

RE: MS: 1121892766193108 – Success with antiretroviral treatment for children in Kigali, Rwanda: experience with health center/nurse-based care

We have done the formatting changes as requested.

1) Figure 1 has been fitted into one page.
2) The figures have been cropped according to the guidelines provided.
2) The underlined text has been changed.

While hoping that these changes would be satisfactory, we remain completely open to further suggestions and comments.

Yours Sincerely,

Johan van Griensven
Médecins sans Frontières (MSF)
6089, Kigali, Rwanda

Tel: (250) 08597623
Fax: (250) 510993
E-mail: jvgrie@yahoo.com